Author's response to reviews

Title: Awareness of Human papillomavirus and factors associated with intention to obtain HPV vaccination among Korean youth: Quasi experimental study

Authors:

Hae Won Kim (haewon@snu.ac.kr)

Version: 5  Date: 19 January 2015

Author's response to reviews: see over
Dear Editor

I am hereby submitting the revision of the above manuscript in the *BMC International Health & Human Rights*. The revision addresses all of the comments of the editor. The comments have significantly improved the manuscript, but please let me know if any further changes are required prior to acceptance.

Thank you for considering the manuscript for publication.

Yours faithfully,

Corresponding author:
Hae Won KIM (RN, PhD)
Associate Professor
The research institute of nursing science, College of Nursing, Seoul National University, Taehakro 103, Jongro-Gu, Seoul, South Korea 110-799
Tel.: +82-2-740-8820; Fax: +82-2-765-4103
Email: haewon@snu.ac.kr

Editor’s comments:
Introduction should be renamed background and there should be a separate conclusion section, funding should be included in an acknowledgment section.

Answer:
According to the comments, the corrections were made in the text.

Thank you for these comments.